Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody-Associated Vasculitis with Severe Kidney Disea...





J Am Soc Nephrol. 2020 Nov; 31(11): 2688–2704. Published online 2020 Aug 21. doi: 10.1681/ASN.2019111197: 10.1681/ASN.2019111197 PMCID: PMC7608964 PMID: <u>32826324</u>

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody– Associated Vasculitis with Severe Kidney Disease

Marta Casal Moura,¹ Maria V. Irazabal,² Alfonso Eirin,² Ladan Zand,² Sanjeev Sethi,³ Bijan J. Borah,^{4,5} Jeffrey L. Winters,⁶ James P. Moriarty,^{4,5} Rodrigo Cartin-Ceba,⁷ Alvise Berti,¹ Misbah Baqir,¹ Gwen E. Thompson,¹ Ashima Makol,⁸ Kenneth J. Warrington,⁸ Viengneesee Thao,^{4,5} Ulrich Specks,^{©1} and Fernando C. Fervenza^{©2}

¹Division of Pulmonary and Critical Care, Department of Medicine, Mayo Clinic College of Medicine and Science, Rochester, Minnesota

²Division of Nephrology and Hypertension, Department of Medicine, Mayo Clinic College of Medicine and Science, Rochester, Minnesota

³Division of Anatomic Pathology, Department of Laboratory Medicine and Pathology, Mayo Clinic College of Medicine and Science, Rochester, Minnesota

⁴Robert D. and Patricia E. Kern Center for the Science of Health Care Delivery, Mayo Clinic, Rochester, Minnesota ⁵Division of Health Care Policy and Research, Department of Health Sciences Research, Mayo Clinic, Rochester, Minnesota

⁶Division of Transfusion Medicine, Department of Laboratory Medicine and Pathology, Mayo Clinic College of Medicine and Science, Rochester, Minnesota

⁷Division of Pulmonary Medicine, Department of Medicine, Mayo Clinic Arizona, Phoenix, Arizona

⁸Division of Rheumatology, Department of Medicine, Mayo Clinic College of Medicine and Science, Rochester,

Minnesota

Corresponding author.

Correspondence: Dr. Ulrich Specks, Division of Pulmonary and Critical Care Medicine, Mayo Clinic College of Medicine, 200 First Street SW, Rochester, MN 55905, or Dr. Fernando C. Fervenza, Division of Nephrology and Hypertension, Mayo Clinic College of Medicine, 200 First Street, SW, Rochester, MN 55901. E-mail: specks.ulrich@mayo.edu or fervenza.fernando@mayo.edu

Received 2019 Nov 21; Accepted 2020 Jul 19.

Copyright © 2020 by the American Society of Nephrology

Significance Statement

Efficacy of rituximab (RTX) in ANCA-associated vasculitis (AAV) in patients with severe renal involvement (eGFR<30 ml/min per 1.73 m²) has not been addressed in clinical trials. This observational study did not find statistically significant differences between RTX and cyclophosphamide (CYC) for remission-induction therapy or any apparent benefit from the addition of plasma exchange (PLEX) to standard remission-induction therapy for patients with AAV and severe renal in-

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody-Associated Vasculitis with Severe Kidney Disea...

volvement. Although our analyses suggest that the benefits and risks of these therapeutic choices (RTX versus CYC with and without PLEX) are balanced, a randomized, controlled trial is needed to confirm these findings.

Keywords: ANCA, glomerulonephritis, rituximab, cyclophosphamide, Plasmapheresis

Visual Abstract

Keywords: ANCA, glomerulonephritis, rituximab, cyclophosphamide, Plasmapheresis

Abstract

Background

Treatment of patients with ANCA-associated vasculitis (AAV) and severe renal involvement is not established. We describe outcomes in response to rituximab (RTX) versus cyclophosphamide (CYC) and plasma exchange (PLEX).

Methods

A retrospective cohort study of MPO- or PR3-ANCA–positive patients with AAV (MPA and GPA) and severe kidney disease (eGFR <30 ml/min per 1.73 m²). Remission, relapse, ESKD and death after remission-induction with CYC or RTX, with or without the use of PLEX, were compared.

Results

Of 467 patients with active renal involvement, 251 had severe kidney disease. Patients received CYC (n=161) or RTX (n=64) for remission-induction, and 51 were also treated with PLEX. Predictors for ESKD and/or death at 18 months were eGFR <15 ml/min per 1.73 m² at diagnosis (IRR 3.09 [95% CI 1.49 to 6.40], P=0.002), renal recovery (IRR 0.27 [95% CI 0.12 to 0.64], P=0.003) and renal remission at 6 months (IRR 0.40 [95% CI 0.18 to 0.90], P=0.027). RTX was comparable to CYC in remission-induction (BVAS/WG=0) at 6 months (IRR 1.37 [95% CI 0.91 to 2.08], P=0.132). Addition of PLEX showed no benefit on remission-induction at 6 months (IRR 0.73 [95% CI 0.44 to 1.22], P=0.230), the rate of ESKD and/or death at 18 months (IRR 1.05 [95% CI 0.51 to 2.18], P=0.891), progression to ESKD (IRR 1.06 [95% CI 0.50 to 2.25], P=0.887), and survival at 24 months (IRR 0.54 [95% CI 0.16 to 1.85], P=0.330).

Conclusions

The apparent benefits and risks of using CYC or RTX for the treatment of patients with AAV and severe kidney disease are balanced. The addition of PLEX to standard remission-induction therapy showed no benefit in our cohort. A randomized controlled trial is the only satisfactory means to evaluate efficacy of remission-induction treatments in AAV with severe renal involvement.

Renal involvement is common in patients with ANCA-associated vasculitis (AAV) (64%–85%), often presenting as rapidly progressive GN with a pauci-immune necrotizing crescentic GN on renal biopsy sample histology,^{1_4} and it is associated with increased morbidity and mortality.⁵ Despite advances in therapy, a significant number of patients progress to ESKD.^{6_8} In addition, outcome data derived from patients with severe kidney disease are scarce and are on the basis of data from small cohorts.²¹⁰

Remission-induction regimens with rituximab (RTX) or cyclophosphamide (CYC) were shown to have similar efficacy and safety in patients with mild-to-moderate renal disease secondary to AAV in the RAVE (Rituximab in ANCA-Associated Vasculitis) trial.^{11_13} However, treatment responses to RTX and outcomes in patients with more advanced degrees of renal dysfunction were only addressed in the RITUXVAS (rituximab versus cyclophosphamide in ANCA-associated vasculitis) trial, which included 33 patients randomized to a combined treatment of RTX and CYC, and 11 patients to CYC only independently of the severity of renal involvement.^{14,15} In that study, the isolated efficacy of RTX was not evaluated as the patients treated with RTX also received two pulses of intravenous (iv) CYC.^{14,15}

The use of plasma exchange (PLEX) therapy for remission-induction of AAV is controversial.¹⁶–³² The scientific rationale for the use of PLEX is the suspected pathogenic role of ANCA.³³ An increase in renal recovery rates at 12 months in patients dependent on dialysis with the addition of PLEX to remission-induction therapy for patients with severe ANCA-associated GN was observed in the MEPEX (methylprednisolone versus plasma exchange) trial studying patients with baseline serum creatinine (SCr) >5.8 mg/dl.¹⁶–¹⁸ However, other studies found no benefit of adding PLEX to standard remission-induction therapy,²⁷–²⁹ or in patients with microscopic polyangiitis (MPA) or eosinophilic granulomatosis with polyangiitis.³⁰ Moreover, long-term follow-up of the MEPEX trial patients showed that the renal survival benefit disappeared after 12 months.²⁰ Furthermore, the recent report of the PEXIVAS (plasma exchange and glucocorticoid dosing in the treatment of ANCA-associated vasculitis) trial showed no benefit from adding PLEX to standard remission-induction therapy in patients with eGFR<50 ml/min per 1.73 m².³⁴

This study was conducted to (1) describe the remission-induction outcomes in response to RTX compared with CYC and to (2) evaluate the benefits of adding PLEX to standard remission-induction therapy, in patients with AAV and severe kidney disease.

Methods

Study Design

After obtaining approval from the Institutional Review Board, we conducted a single-center retrospective cohort study of all consecutive patients with AAV-associated renal disease evaluated at Mayo Clinic from January 1, 1996 to December 31, 2015 with last follow-up on December 31, 2017. The "Mayo Clinic ANCA-associated vasculitis cohort" comprised a total of 1830 patients with

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody–Associated Vasculitis with Severe Kidney Disea...

a diagnosis of AAV verified by vasculitis experts. Patients with renal involvement were identified by applying an internal search engine (Advanced Cohort Explorer) using the following keywords: "kidney disease," "renal failure," "pauci-immune GN," "hematuria," "proteinuria," and "renal biopsy."

Patient Characteristics

All data were abstracted retrospectively from electronic medical records and included demographic characteristics, comorbidities, laboratory findings, biopsy results, therapies, and outcomes. The date of diagnosis of severe renal involvement (index date) was registered for the calculation of outcome time-points. The Birmingham Vasculitis Activity Score for Wegener Granulomatosis (BVAS/WG) was used to quantify disease activity at presentation and during the follow-up.^{35_37}

Myeloperoxidase (MPO)- or proteinase 3 (PR3)–ANCA–positive patients with newly diagnosed AAV or relapsing disease with active renal involvement and fulfilling the American College of Rheumatology criteria and Chapel Hill consensus definition for granulomatosis with polyangiitis and MPA were included (Figure 1).^{38–40} Patients with positive anti-GBM antibodies, eosinophilic granulomatosis with polyangiitis, AAV without evidence of GN, or who were MPO- or PR3-ANCA negative were excluded. Patients with at least one follow-up visit after their index date were included. Active renal involvement was defined by the presence of (1) active, biopsy-proven, pauci-immune GN; (2) red blood cell casts on urine microscopy; or (3) rise in SCr >30% (or >25% decline in creatinine clearance) attributed to active vasculitis. Severe kidney disease was defined by eGFR<30 ml/min per 1.73 m² at diagnosis of renal disease secondary to AAV. Patients were grouped according to the interventions received for remission-induction (CYC or RTX and with or without PLEX). No patient received RTX in combination with CYC. For the purpose of this study, which was to evaluate different treatment options in patients with AAV and severe kidney disease, time of diagnosis of severe renal involvement coincides with the start of the remission-induction treatments.

Renal Function Assessment

eGFR was calculated using the Chronic Kidney Disease Epidemiology Collaboration equation, $\frac{41.42}{7}$ and was recorded at baseline and at 6, 12, 18, and 24 months. Further categorization of severity of renal impairment at baseline for the determination of predictors of outcomes included the eGFR cut-off (<15 and ≥15–30 ml/min per 1.73 m²). Time-trend values of eGFR were determined for 5-year periods and compared between groups.

Remission-Induction Therapies and PLEX

Indication for remission-induction and treatment used were not defined by a pre-established protocol but decided by the treating physicians. Patients who received CYC (oral, 2 mg/d for 6 months) followed a strict clinical practice protocol of laboratory monitoring, every 1–2 weeks, for early detection of leukopenia. Patients who received RTX (iv, 375 mg per m² of body surface area once weekly for 4 weeks) had absolute B-cell counts measured at intervals determined by the

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody-Associated Vasculitis with Severe Kidney Disea...

treating physician. The tapering of glucocorticoids was not performed following a pre-established protocol. The decision to use PLEX was also determined by the treating physician. The replacement fluid used for PLEX at our institution was 5% albumin with 3 U of fresh frozen plasma as the final portion in the presence of bleeding in accordance with the American Society of Apheresis guidelines.⁴³ The apheresis devices used during the study period to perform the PLEX procedures were the TerumoBCT COBE Spectra, the TerumoBCT Spectra Optia, and the Fresenius KABI Fenwal Amicus. Anticoagulant consisted of acid citrate dextrose solution A with the addition of 10,000 U of heparin in patients without bleeding risk and acid citrate dextrose solution A only in those with bleeding risk. A 1-plasma-volume exchange was performed daily for 7–14 days with the apheresis devices calculating the volume to be exchanged on the basis of a hematocrit obtained within 24 hours of the procedure.⁴³ As per our practice, RTX was not given within the 48 hours before receiving a PLEX treatment.

Outcomes Assessment

Remission was defined by a BVAS/WG of 0 independent of the dose of prednisone, and complete remission was defined by a BVAS/WG of 0 with a complete discontinuation of prednisone. These events were assessed at 6 months and during follow-up. Renal remission was defined as improvement of hematuria (<10 red cells per high-power field), and improvement (eGFR increase >10%) or stabilization (stable eGFR or decrease <15%) of renal function. Relapse was defined by an increase of BVAS/WG >1 that resulted in therapy changes (increases in doses of maintenance-remission therapy or the start of a new remission-induction cycle). The number of relapses after achievement of remission, the type of relapse (major or minor), the organ involvement (renal versus nonrenal), and the BVAS/WG at the time of relapse were recorded.

Kidney disease progression was determined by the development of ESKD defined as eGFR<15 ml/min per 1.73 m² or the need to initiate RRT. Renal recovery was defined as independence of RRT for those in whom this therapy was initiated. Date of death was recorded to assess survival. Remission at 6 months and "combined events" end point of ESKD and/or death at 18 months were defined as the primary outcomes of the study. Survival, ESKD, and combined events at 24 months were secondary outcomes.

Statistical Analyses

Categoric variables were presented as count (percentage), whereas continuous variables were presented as mean (SD) if they were normally distributed as determined by Shapiro–Wilk test, or as median (interquartile range) if non-normal. For comparisons of categoric variables between groups, the Pearson's chi-squared test was used if the number of elements in each cell was ≥ 5 ; Fisher's exact test was used otherwise. For comparison of continuous variables between groups, an unpaired *t* test for independent samples was used for distributions consistent with normality, and the Mann–Whitney *U* test was used otherwise. We dichotomized age into <60 years and ≥ 60 years to include the variable in the logistic regression for the estimation of risk factors for severe renal involvement, as the same cut-off was used in the recent PEXIVAS trial.³⁴ We used the receiver operated curves to confirm whether this cut-off fit our data. eGFR was dichotomized according with pre-established classification cut-offs used in clinical practice.⁴⁴

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody-Associated Vasculitis with Severe Kidney Disea...

Logistic regression models were developed to examine the predictive role of the baseline clinical characteristics for the development of severe kidney disease. Variables were considered for the multivariate logistic regression models if they occurred before the development of the outcome of interest, had <10% of missing values, had *P* values <0.05 in the univariate analysis, and were clinically plausible. The final model was determined using both clinical and statistical criteria, taking into consideration collinearity, interaction, and the number of patients who experienced the outcome of interest. Some of the continuous variables were categorized with cut-offs determined according to pre-established guidelines or clinical practice.⁴⁴ The odds ratios (ORs) with 95% confidence intervals (95% CIs) were reported when appropriate.

The Kaplan–Meier method was used to assess cumulative incidence of remission, time to relapse, cumulative incidence of ESKD, time to death (survival), and cumulative incidence of combined events of ESKD and/or death at the more relevant time points. Cox proportional hazards regression models were used to determine predictive factors of the outcomes. We report the incidence rate ratio (IRR) with a 95% CI when appropriate.⁴⁵ We treated the patient's observation as right-censored: we included the observation in the survival analysis up to the last point at which the outcome was known to have not yet occurred. A multivariable Cox proportional hazards regression model was used to assess the effect of being treated in each decade on the probability of the outcome.

In addition, we also performed propensity score (PS) matching analysis with the objective to match patients by severity and to account for potential unequal distribution of important covariates between groups resulting from potential nonrandom assignment typical in observational studies like ours. The PS or the probability of receiving CYC versus RTX and that of receiving PLEX versus no PLEX were calculated separately for the two comparisons using logistic regression models. The number of covariates used in the PS models was conditioned by the number of patients assigned to RTX and PLEX. For the probability of being assigned to CYC versus RTX, the PS model included the following variables: eGFR<15 ml/min per 1.73 m² at renal involvement secondary to AAV diagnosis, alveolar hemorrhage, treatment with PLEX, and use of iv methylprednisolone as part of the remission-induction protocol. Similarly, the PS model for PLEX versus no PLEX included the following variables: eGFR<15 ml/min per 1.73 m², alveolar hemorrhage, treatment with CYC versus RTX, and use of iv methylprednisolone as part of the remission-induction protocol. We applied nearest-neighbor PS matching without replacement with a caliper of 0.001. After matching, effects on binary (yes/no) outcomes were assessed through ORs, and effects on time-toevent outcomes through IRRs. Model fit calibration was assessed by the Hosmer-Lemeshow goodness-of-fit test. We verified the performance of the PS matching by comparing the balance in the distribution of the variables between groups pre- and post-PS matching. *P* values <0.05 (twosided) were considered significant. We did not adjust for the matched pairs arising from the PS matching.^{46,47} Finally, after generating a dichotomous variable for the decade, we built a multivariable logistic regression model to assess the effect of being treated for each decade on the probability of the outcome after PS matching analysis.

IBM SPSS Statistics for MacOS, version 25 (IBM, Armonk, NY) was used for all data analysis with exception of the PS matching analysis that was calculated using Stata, StataCorp, version 13.1 (College Station, TX).

Results

Patient Characteristics and Clinical Outcomes

Of the 1830 patients with AAV evaluated during the study period, active renal disease was documented in 467 (25.5%) whereas 251 (13.7%) met the inclusion criteria of severe kidney disease (Figure 1). Baseline demographics and outcomes for the 251 patients with severe kidney disease are presented in Tables 1 and 2, respectively. Dialysis was required in 54 (21.5%) patients, 38 (72.2%) within 4 weeks of renal involvement diagnosis, and 12 (22.2%) of those were subsequently able to discontinue RRT: nine patients treated with CYC and three patients treated with RTX, and in seven patients PLEX was added to remission-induction immunosuppression. Remission (BVAS/WG=0) was achieved in 168 (66.9%) patients at 6 months and in a total of 198 (78.8%) patients during the follow-up. Renal remission was observed in 177 (70.5%) patients at 6 months and in a total of 199 (79.3%) patients during the follow-up. Relapses were documented in 81 (32.3%) patients, and in 46 (18.3%) patients these were renal relapses. PR3-ANCA-positive patients had a higher proportion of relapses in comparison with MPO-ANCA-positive patients (43.7% versus 34.1%, *P*=0.158). Nine patients died within 4 weeks of presentation, and a total of 13 patients died before 6 months (eight in the CYC group versus five in the RTX group, P=0.419). At the end of the follow-up (median of 5.6 years [interquartile range, 1.9–11.5]), a total of 62 patients (24.7%) had died, with death attributed to active AAV in 19 patients.

In patients who achieved remission, eGFR improved in the first 6 months, and continued to improve until 24 months (Figure 2), irrespective of the therapy used for remission-induction. At 6 months, the proportion of patients that were ANCA-negative was comparable for CYC (28.0%) and RTX (20.3%, P=0.231), and with or without the addition of PLEX (25.0% versus 23.5%, P=0.683).

Predictors of Severe Kidney Disease

Compared with patients with active nonsevere renal involvement (n=216) (for demographic characteristics please refer to <u>Supplemental Table 1</u>), patients with severe kidney disease at diagnosis (n=251) were more likely to be older than 60 years of age (OR 2.27; 95% CI, 1.60 to 3.29; P<0.001) and to have arterial hypertension (OR 2.13; 95% CI, 1.44 to 3.15; P<0.001), diabetes mellitus (OR 2.67; 95% CI, 1.56 to 4.57; P<0.001), dyslipidemia (OR 1.79; 95% CI, 1.20 to 2.69; P=0.005), body mass index (BMI) ≥30 kg/m² (OR 1.52; 95% CI, 1.01 to 2.28; P=0.046), renal limited disease (OR 3.68; 95% CI, 2.26 to 5.98; P<0.001), MPA (OR 1.84; 95% CI, 1.27 to 2.65; P=0.001), MPO-ANCA (OR 2.09; 95% CI, 1.44 to 3.03; P<0.001), and lower levels of hemoglobin (OR 1.53; 95% CI, 1.38 to 1.75; P<0.001). Multivariable logistic regression analysis showed that older age, arterial hypertension, diabetes mellitus, BMI≥30 kg/m², renal limited disease, and lower levels of hemoglobin, but not MPO-ANCA, were independently associated with the severity of renal disease at diagnosis (<u>Supplemental Table 2</u>).

RTX versus CYC for Remission-Induction

Of the 251 patients with severe renal involvement, CYC was the remission-induction immunosuppressant of choice in 161 (64.1%) patients and RTX in 64 (25.5%) patients, whereas 26 patients were treated with other therapies (Figure 1). All patients received glucocorticoids at the time of remission-induction. The combination of iv methylprednisolone boluses with either CYC or RTX (59.0% versus 81.3%) followed by oral glucocorticoids was the main strategy (Table 3). Patients treated with CYC had lower mean eGFR at diagnosis in comparison with patients treated with RTX (15.6 versus 17.7 ml/min per 1.73 m², P=0.050) (Figure 2A). This was also observed in patients who received PLEX and CYC in comparison with those who received PLEX and RTX (12.5 versus 18.5 ml/min per 1.73 m², P=0.026) (Figure 2C).

There were no statistically significant differences between CYC and RTX in the frequency of remission (BVAS/WG = 0) at 6 months (76.6% versus 83.3%, P=0.291), combined events of ESKD and/or death at 18 months (29.8% versus 23.4%, P=0.336), death at 24 months (8.1% versus 12.5%, P=0.303), ESKD at 24 months (28.0% versus 15.6%, P=0.052), and combined events of ESKD and/or death at 24 months (31.1% versus 23.4%, P=0.255) (Table 4). In contrast, in comparison with CYC, complete remission (BVAS/WG=0 without prednisone) at 6 months was more frequently achieved in the RTX group (17.7% versus 31.7%, P=0.031). In comparison with CYC, remission-induction treatment with RTX allowed faster prednisone tapering and subsequent discontinuation (median prednisone dose at 6 months, 10.0 versus 2.5 mg, P=0.002; 12 months, 5.0 versus 0.0 mg, P=0.001; 18 months, 3.0 versus 0.0 mg, P=0.036) (Table 3). Renal recovery occurred at the same pace in both treatment groups over 6 months despite the baseline differences in eGFR (Figure 2A).

Using the unadjusted Kaplan–Meier method, we determined that time to remission at 6 months was different between CYC and RTX (4.2 versus 3.5 months, *P*=0.002), but the time to combined events at 18 months was not statistically significantly different between CYC and RTX groups (<u>Supplemental Figure 1</u>).

To adjust for observed imbalances in the distribution of disease features including the severity of renal impairment and treatments, we performed PS analysis for matching the patients according to the presence of eGFR<15 ml/min per 1.73 m^2 versus $\geq 15-30 \text{ ml/min per } 1.73 \text{ m}^2$ at diagnosis, alveolar hemorrhage, administration of iv methylprednisolone as part of the remission-induction treatment, and receiving PLEX or no PLEX (Supplemental Table 3). The incidences of the outcomes in the matched sample indicate no statistically significant differences between groups (Supplemental Table 4). Patients treated with RTX showed a statistically nonsignificantly shorter time to achieve remission (BVAS/WG=0) (time to event estimated by post-PS Kaplan–Meier analysis, 3.5 versus 4.1 months, P=0.069) at 6 months (Figure 3A) and a statistically nonsignificantly higher probability of achieving remission (BVAS/WG=0) by 6 months (IRR 1.37; 95% CI, 0.91 to 2.08; P=0.132) (Table 5) compared with patients treated with CYC. There were no statistically significant differences in the probability or time to relapse at 12 months after remission, combined events of ESKD and/or death at 18 months (Figure 3B), survival at 24 months, progression to ESKD at 24 months, and combined events of ESKD and/or death at 24 months (Table 5). In addition, in order to demonstrate our rationale on the renal involvement severity cut-off choice, we

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody-Associated Vasculitis with Severe Kidney Disea...

also provide the comparison of clinical characteristics and outcomes between the two groups with eGFR<15 ml/min per 1.73 m² versus \geq 15–30 ml/min per 1.73 m² at diagnosis in <u>Supplemental</u> <u>Table 5</u>.

PLEX as an Adjunct to Remission-Induction Therapy

PLEX was added to standard immunosuppressive remission-induction therapy in 51 (20.3%) patients. The underlying remission-induction immunosuppressant was CYC in 37 (72.5%) and RTX in 14 (27.5%). Patients who received PLEX also received concomitant iv methylprednisolone more frequently (90.2% versus 55.5%, P<0.001) (Table 6). Indications for PLEX were severe renal involvement in 48 (94.1%) patients and renal involvement with alveolar hemorrhage in 19 (37.3%) patients (Table 1). The median number of PLEX sessions was 7. Patients who received PLEX presented more frequently with eGFR<15 ml/min per 1.73 m² (56.9% versus 37.5%, P=0.012), had lower mean eGFR at diagnosis (14.1 versus 16.8 ml/min per 1.73 m², P=0.035), required renal function replacement therapy more frequently (39.3% versus 17.0%, P<0.001) (Figure 2B), and had higher BVAS/WG at diagnosis (9 versus 7 points, P=0.007). From the 222 patients with biopsy-proven diagnosis of AAV with severe renal involvement, we were able to review 199 biopsy samples (90% of the cases). The histologic pattern was focal in 39 (19.6%), crescentic in 49 (24.6%), mixed in 73 (36.7%), and sclerotic in 38 (19.1%) patients. There were no statistically significant differences in the distribution of the histologic categories between patients that received PLEX in comparison with the patients who did not receive PLEX (P=0.375).

In the unadjusted analysis, there were no statistically significant differences between patients who did not receive PLEX in comparison with those who did in the achievement of remission (BVAS/WG=0) at 6 months (78.1% versus 73.5%, P=0.497), complete remission (BVAS/WG=0 without prednisone) at 6 months (21.7% versus 23.9%, P=0.754), or renal remission at 6 months (79.4% versus 77.6%, P=0.775); or in the combined events of ESKD and/or death at 18 months (24.5% versus 35.3%, P=0.120), or death at 24 months (9.0% versus 7.8%, P=0.794) (Table 7). However, among patients treated with PLEX, a statistically nonsignificantly higher incidence of ESKD was observed at 24 months compared with patients not treated with PLEX (33.3% versus 21.0%, P=0.064). The pace of prednisone tapering was not statistically significantly different between patients treated with or without PLEX (median prednisone dose at 6 months, 7.5 versus 5.0 mg, P=0.386; 12 months, 5.0 versus 5.0 mg, P=0.927; 18 months, 0.0 versus 0.0 mg, P=0.289). The frequency of relapses by 12 months after remission (14.9% versus 13.1%, P=0.745) was not statistically significantly different between groups. Renal recovery occurred at similar rates in both groups over 6 months despite the baseline difference in eGFR observed (Figure 2B), and this was independent of the choice of remission-induction therapy (Figure 2C).

Using the unadjusted Kaplan–Meier method, we observed that there was no statistically significant difference of adding PLEX on the achievement of remission (BVAS/WG=0) at 6 months, relapse at 12 months, and combined events of ESKD and/or death at 24 months. However, combined events of ESKD and/or death at 18 months tended to occur earlier in patients who received PLEX (13.6 versus 14.4 months, *P*=0.099) (<u>Supplemental Figure 2</u>). When the same analysis was performed

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody-Associated Vasculitis with Severe Kidney Disea...

only in patients that received PLEX, and stratified by immunosuppression therapy received for remission-induction, we observed no statistically significant differences in the time to the outcomes between the combination of PLEX with CYC or RTX (<u>Supplemental Figure 3</u>).

To adjust for the confounding introduced by differences in disease severity in patients who received PLEX compared with those who did not, PS analysis was performed using the previously mentioned variables. In this case, the matching accounted also for the probability of receiving CYC or RTX (<u>Supplemental Table 6</u>). The outcome incidence after PS matching is displayed in <u>Supplemental Table 7</u>, which showed no statistically significant difference between the two groups. There were no statistically significant differences in the probability and time to remission (BVAS/WG=0) at 6 months (Figure 4A), relapse at 12 months after remission, combined events of ESKD and/or death at 18 months (Figure 4B), survival at 24 months, progression to ESKD at 24 months, and combined events of ESKD and/or death at 24 months (<u>Table 5</u>). After PS matching, we confirmed that adding PLEX had no benefit in the rate of improvement of eGFR (<u>Supplemental Figure 4</u>).

Predictors of Combined Events of ESKD and/or Death at 18 Months

Patients who incurred the combined end point of ESKD and/or death by 18 months more often remained ANCA-positive at 12 months after remission-induction treatment than those who did not (IRR 3.30; 95% CI, 1.05 to 10.37; P=0.041), had an eGFR<15 ml/min per 1.73 m² at diagnosis (IRR 3.00; 95% CI, 1.82 to 4.95; P<0.001), and had lower eGFR at 6 and 12 months (respectively, IRR 1.10; 95% CI, 1.07 to 1.14; P<0.001; IRR 1.07; 95% CI, 1.04 to 1.11; P<0.001). Renal function recovery for an eGFR>30 ml/min per 1.73 m² at 6 months and renal remission at 6 months were protective factors for the achievement of the end point (respectively, IRR 0.25; 95% CI, 0.11 to 0.57; P=0.001; and IRR 0.25; 95% CI, 0.15 to 0.42; P<0.001). Multivariable Cox regression showed that patients with eGFR<15 ml/min per 1.73 m² at diagnosis were at higher risk to evolve to ESKD and/or death at 18 months and that renal recovery and achieving renal remission at 6 months were protective factors for this outcome (<u>Supplemental Table 8</u>).

Outcome Comparison per Decade

In order to assess any potential time-trend bias, we adjusted for decades while evaluating outcomes in the propensity-matched sample. There were no statistically significant differences in the risks for the outcomes between decades (<u>Supplemental Table 9</u>).

Discussion

Our study showed that after using PS matching analysis for adjustment for confounding introduced by different disease severity, remission (BVAS/WG=0) at 6 months was not statistically significantly different, whether CYC or RTX was used for induction of remission, or whether PLEX was added as an adjunct therapy in patients with severe kidney disease. In accordance with previous studies, renal function at presentation was the main determinant of ESKD-free survival or death in patients with renal involvement of AAV.^{48,49} In our cohort, patients presented with a mean

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody-Associated Vasculitis with Severe Kidney Disea...

eGFR value of 16.3 ml/min per 1.73 m², similar to the values reported in other smaller cohorts.^{9,10} The presence of MPO-ANCA (versus PR3-ANCA) has previously been reported as a determinant of renal outcome, ^{5,50,51} but multivariable modeling did not identify it as a risk factor for the development of severe kidney disease in our cohort. Moreover, we showed that the presence of classic cardiovascular risk factors and increased age appear to aggravate the susceptibility of the kidney for severe kidney disease in ANCA-associated GN.

The RAVE and RITUXVAS trials provided robust data about the efficacy of RTX for remission-induction in comparison with CYC.^{11,14} In the RAVE trial, patients with an SCr>4 mg/dl at baseline were excluded. Nevertheless, eGFR was lower than 30.0 ml/min per 1.73 m² in 35% of patients randomized to the RTX group and in 27% of those randomized to the CYC group.^{11,13} Similarly, in the RITUXVAS trial, the median eGFR was 20.0 ml/min per 1.73 m² in the RTX arm and 12.0 ml/min per 1.73 m² in the CYC arm.¹⁴ A *post-hoc* analysis of the RAVE trial found that patients enrolled with an eGFR<30.0 ml/min per 1.73 m² at baseline responded similarly to RTX and CYC, and there was no statistically significant difference in the mean eGFR increase over the 18 months of follow-up.¹³ In this study, which includes the largest number of patients with severe kidney disease treated with RTX (*n*=64), remission (BVAS/WG=0) at 6 months was achieved at similar rates in patients in the group treated with CYC compared with patients in the group treated with RTX, which is in line with the results reported in the RAVE trial.^{11,13}

After PS matching analysis, RTX remained, at a minimum, not statistically significantly different from CYC as remission-induction therapy in patients that presented with severe kidney disease in AAV. Furthermore, the group treated with RTX achieved remission earlier, which might have contributed to the improvement of renal outcomes and renal recovery and be related to the lower incidences of overall ESKD and combined events in this group.

PLEX has been advocated as an adjunct therapy in AAV with renal involvement but without a clear consensus.^{16_32} The MEPEX trial included patients with severe kidney disease (median SCr of 8.31 mg/dl) and, on the basis of the results, it was concluded that PLEX was of initial benefit for patients with severe renal failure,¹⁸ but the effect was lost over longer term follow up.²⁰ However, results of the PEXIVAS trial in patients with newly diagnosed or relapsing severe AAV, including those with alveolar hemorrhage and/or GN (eGFR<50 ml/min per 1.73 m²), who were randomized to receive or not receive PLEX for up to seven sessions during the span of 14 days as an adjunct to standard remission-induction therapy, indicate that PLEX does not reduce the risk of ESKD or death in patients with AAV.³⁴ This was even true when the combined outcome of ESKD or death was evaluated at 12 months and it is consistent with our observations where the addition of PLEX to standard remission-induction therapy did not improve the achievement of remission or reduce the progression to ESKD and/or death in patients with AAV with severe kidney disease. Furthermore, a prior study from our group showed that in patients with AAV and diffuse alveolar hemorrhage addition of PLEX as an adjunct to standard remission-induction therapy to their treatment.⁵² Consequently, we found no evidence to support the use of PLEX as an adjunct to standard remission-induction therapy in AAV.

This study has limitations inherent to its retrospective design. First, our cohort consists of a Midwestern United States white population with predominantly Scandinavian and Northern European ethnic backgrounds; therefore, the results may not be generalizable. Second, the stan-

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody-Associated Vasculitis with Severe Kidney Disea...

dard of care between 1996 and 2010 was to use CYC, with RTX use becoming widespread only after the publication of the RAVE¹¹ and RITUXVAS trials in 2010,¹⁴ and therefore a time-trend bias regarding the prescription of RTX versus CYC was considered. However, the differences in the standard of care between decades did not translate into differences in the risk for the outcome estimation after PS matching analysis. Third, treatment assignments were not protocol-driven and thus prone to selection biases on the basis of the disease severity and degree of renal impairment. For this reason, we adjusted our analyses for confounding variables using PS matching analysis as previously reported by our group.⁵² Fourth, the dosing and tapering of glucocorticoids was not protocolized, and complete discontinuation of prednisone by 6 months was not a primary intent in the majority of cases. Therefore, it is no surprise that the primary outcome of the RAVE trial (complete remission at 6 months) was met by fewer patients in our study. For similar reasons, our outcome of remission at 6 months cannot directly be compared with the secondary end point of the RAVE trial of BVAS/WG of 0 and prednisone dose of <10 mg at 6 months, even though it comes close as the median prednisone dose at 6 months was 10 mg. $\frac{11}{7}$ Lastly, we cannot completely exclude the possibility that different maintenance therapy choices may have had minor variable effects on the various outcomes evaluated in our study.

Despite these limitations, this is the largest cohort of patients with active severe renal involvement of AAV providing a detailed analysis of clinical characteristics and outcomes in response to different treatments in real clinical practice. Even though the treatment of these patients does not follow a strictly standardized protocol, there is homogeneity in the care of these patients by a group of experts in AAV, who followed consistent patterns and decisions. In addition, the high frequency of biopsy-proven severe kidney disease makes this large patient population the best-characterized cohort of patients with AAV with severe renal involvement reported to date.

Our observational study fills important gaps left by various clinical trials. The population included in our cohort comprises a spectrum of patients ranging from patients similar to those enrolled in the MEPEX and RITUXVAS trials to the patients with severe kidney disease (eGFR<30 ml/min per m²) enrolled in the RAVE and PEXIVAS trials. This is the only study that compares the outcomes of RTX versus CYC for remission-induction in such a large number of patients with this spectrum of renal disease, while at the same time evaluating the effects of PLEX.^{11,4,18,34} Moreover, our study adds to the PEXIVAS trial results in three important ways: (1) by adding a large population of patients with eGFR<30 ml/min per 1.73 m² (*n*=251); (2) we used a more aggressive approach to PLEX because, as per our practice, PLEX was generally given for a minimum of seven sessions and up to 14 session, versus PEXIVAS trial sites that qualified for enrollment in PEXIVAS were enrolled in the trial, whereas all patients that were seen at our institution, and that we could ascertain an outcome for, were included. As such, our study represents a more faithful picture of real practice in patients with AAV and severe renal and pulmonary disease.

In conclusion, our results do not support the preferential use of CYC over RTX or the addition of PLEX to standard remission-induction therapy in patients with AAV and severe kidney disease. However, a randomized trial is the only satisfactory means to evaluate efficacy of remission-induction treatments in patients with AAV and severe renal involvement. The degree of renal dysfunction at diagnosis is the best predictor of ESKD and/or death in patients with severe kidney disease

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody–Associated Vasculitis with Severe Kidney Disea...

in AAV and the achievement of renal remission was the best predictive factor for favorable outcomes. Early diagnosis of renal involvement is crucial in order to prevent irreversible renal damage.

Disclosures

All authors have nothing to disclose.

Funding

The study was supported by an internal grant from the Mayo Nephrology Collaborative Group, Mayo Foundation.

Supplementary Material

Supplemental Data:

Acknowledgments

Dr. Marta Casal Moura, Dr. Ulrich Specks, and Dr. Fernando C. Fervenza designed the study. Dr. Marta Casal Moura, Dr. Maria V. Irazabal, and Dr. Alfonso Eirin abstracted the data. Dr. Marta Casal Moura, Dr. Sanjeev Sethi, Dr. Ulrich Specks, and Dr. Fernando C. Fervenza analyzed the data. Dr. Marta Casal Moura, Dr. Bijan J. Borah, Mr. James P. Moriarty, and Miss Viengneesee Thao performed the statistical analysis. Dr. Marta Casal Moura, Dr. Ladan Zand, Dr. Ulrich Specks, and Dr. Fernando C. Fervenza drafted and revised the paper. All authors provided input for the final version of the manuscript. Dr. Fernando C. Fervenza received unrestricted grants from Genentech Inc., South San Francisco, CA, outside the submitted work. Dr. Ulrich Specks reports grants from Genentech, grants from Bristol Myer Squibb, grants and personal fees from ChemoCentryx, grants from GSK, grants and other from AstraZeneca, and other from InsMed, outside the submitted work. Dr. Kenneth Warrington reports grants from Eli Lilly, grants from Roche/Genentech, grants from GlaxoSmithKline, grants from Kiniksa, personal fees from Roche/Genentech, and personal fees from Sanofi, outside the submitted work.

Footnotes

Published online ahead of print. Publication date available at <u>www.jasn.org</u>.

Supplemental Material

This article contains the following supplemental material online at http://jasn.asnjournals.org/lookup/suppl/doi:10.1681/ASN.2019111197/-/DCSupplemental.

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody-Associated Vasculitis with Severe Kidney Disea...

<u>Supplemental Table 1</u>. Demographic and clinical characteristics description of patients with Anti-Neutrophil Cytoplasmic Antibodies (ANCA) associated vasculitis (AAV) and non-severe kidney disease.

<u>Supplemental Table 2</u>. Multivariable analysis of predictive factors for severe kidney disease in patients with Anti-Neutrophil Cytoplasmic Antibodies (ANCA) associated vasculitis (AAV).

<u>Supplemental Table 3</u>. Analysis of the distribution of the variables included in the propensity score matching analysis in patients with AAV and active severe kidney disease treated with cyclophosphamide (CYC) or rituximab (RTX) for remission induction.

<u>Supplemental Table 4</u>. Post propensity score matching outcomes comparison of patients with AAV and active severe kidney disease treated with cyclophosphamide (CYC) or rituximab (RTX) for remission induction.

<u>Supplemental Table 5</u>. Clinical characteristics, and outcomes of patients with AAV and active severe kidney disease treated stratified according to eGFR at admission: $<15 \text{ mL/min}/1.73\text{m}^2 \text{ vs.} \ge 15 \text{ to} 30 \text{ mL/min}/1.73\text{m}^2$.

<u>Supplemental Table 6</u>. Analysis of the distribution of the variables included in the propensity score matching analysis in patients with AAV and active severe kidney disease according with the status of treatment with plasma exchange (PLEX).

<u>Supplemental Table 7</u>. Post propensity score matching outcomes comparison of patients with AAV and active severe kidney disease according with the status of treatment with plasma exchange (PLEX).

<u>Supplemental Table 8</u>. Multivariable analysis of predictive factors for combined events of ESKD and/or Death at 18 months in patients with Anti-Neutrophil Cytoplasmic Antibodies (ANCA) associated vasculitis (AAV) with active severe Kidney disease.

<u>Supplemental Table 9</u>. Risk for the outcome stratified by treatment after propensity score matching analysis and adjusted per decade (1996–2005 vs. 2006–2015).

<u>Supplemental Figure 1</u>. Remission and combined events before PS matching analysis. Before propensity score matching Kaplan Meier plots of remission achieved over 6 months after initiating remission-induction therapy (1A), and combined events (ESRD and/or death) at 18 months (1B), according with the remission-induction immunosuppression-CYC vs. RTX and before propensity score matching analysis.

<u>Supplemental Figure 2</u>. Remission and combined events before PS matching analysis. Before propensity score matching Kaplan Meier plots of remission achieved over 6 months after initiating remission-induction therapy (2A), and combined events (ESRD and/or death) at 18 months (2B), according with the status of treatment or no treatment with PLEX and before propensity score matching analysis

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody–Associated Vasculitis with Severe Kidney Disea...

<u>Supplemental Figure 3</u>. Remission and combined events before PS matching analysis. Before propensity score matching Kaplan Meier plots of remission achieved over 6 months after initiating remission-induction therapy (3A), and combined events (ESRD and/or death) at 18 months (3B), for patients treated with PLEX stratified by remission-induction immunosuppression-CYC vs. RTX.

<u>Supplemental Figure 4</u>. Changes in mean eGFR with follow up after PS matching analysis. eGFR at baseline, and 6, 12, 18, and 24 months post-diagnosis for severe renal involvement in after PS matching.

References

1. Fauci AS, Haynes BF, Katz P, Wolff SM: Wegener's granulomatosis: Prospective clinical and therapeutic experience with 85 patients for 21 years. *Ann Intern Med* 98: 76–85, 1983. [PubMed: 6336643]

2. Hoffman GS, Kerr GS, Leavitt RY, Hallahan CW, Lebovics RS, Travis WD, et al.: Wegener granulomatosis: An analysis of 158 patients. *Ann Intern Med* 116: 488–498, 1992. [PubMed: 1739240]

3. Berti A, Cornec D, Crowson CS, Specks U, Matteson EL: The epidemiology of antineutrophil cytoplasmic autoantibodyassociated vasculitis in Olmsted county, Minnesota: A twenty-year US population-based study. *Arthritis Rheumatol* 69: 2338–2350, 2017. [PMCID: PMC5711593] [PubMed: 28881446]

4. Jennette JC: Rapidly progressive crescentic glomerulonephritis. *Kidney Int* 63: 1164–1177, 2003. [PubMed: 12631105]

5. de Joode AAE, Sanders JSF, Stegeman CA: Renal survival in proteinase 3 and myeloperoxidase ANCA-associated systemic vasculitis. *Clin J Am Soc Nephrol* 8: 1709–1717, 2013. [PMCID: PMC3789333] [PubMed: 23846462]

6. Booth AD, Almond MK, Burns A, Ellis P, Gaskin G, Neild GH, et al.; Pan-Thames Renal Research Group : Outcome of ANCA-associated renal vasculitis: A 5-year retrospective study. *Am J Kidney Dis* 41: 776–784, 2003. [PubMed: 12666064]

7. Li ZY, Gou SJ, Chen M, Zhao MH: Predictors for outcomes in patients with severe ANCA-associated glomerulonephritis who were dialysis-dependent at presentation: A study of 89 cases in a single Chinese center. *Semin Arthritis Rheum* 42: 515–521, 2013. [PubMed: 23332902]

8. Lee T, Gasim A, Derebail VK, Chung Y, McGregor JG, Lionaki S, et al.: Predictors of treatment outcomes in ANCA-associated vasculitis with severe kidney failure. *Clin J Am Soc Nephrol* 9: 905–913, 2014. [PMCID: PMC4011446] [PubMed: 24578329]

9. Geetha D, Hruskova Z, Segelmark M, Hogan J, Morgan MD, Cavero T, et al.: Rituximab for treatment of severe Kidney disease in ANCA associated vasculitis. *J Nephrol* 29: 195–201, 2016. [PubMed: 25986390]

10. Shah S, Hruskova Z, Segelmark M, Morgan MD, Hogan J, Lee SK, et al.: Treatment of severe Kidney disease in ANCA positive and negative small vessel vasculitis with rituximab. *Am J Nephrol* 41: 296–301, 2015. [PubMed: 26044574]

11. Stone JH, Merkel PA, Spiera R, Seo P, Langford CA, Hoffman GS, et al.; RAVE-ITN Research Group : Rituximab versus cyclophosphamide for ANCA-associated vasculitis. *N Engl J Med* 363: 221–232, 2010. [PMCID: PMC3137658] [PubMed: 20647199]

12. Specks U, Merkel PA, Seo P, Spiera R, Langford CA, Hoffman GS, et al.; RAVE-ITN Research Group : Efficacy of remissioninduction regimens for ANCA-associated vasculitis. *N Engl J Med* 369: 417–427, 2013. [PMCID: PMC5953195] [PubMed: 23902481]

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody-Associated Vasculitis with Severe Kidney Disea...

13. Geetha D, Specks U, Stone JH, Merkel PA, Seo P, Spiera R, et al.; Rituximab for ANCA-Associated Vasculitis Immune Tolerance Network Research Group : Rituximab versus cyclophosphamide for ANCA-associated vasculitis with renal involvement. *J Am Soc Nephrol* 26: 976–985, 2015. [PMCID: PMC4378104] [PubMed: 25381429]

14. Jones RB, Tervaert JW, Hauser T, Luqmani R, Morgan MD, Peh CA, et al.; European Vasculitis Study Group : Rituximab versus cyclophosphamide in ANCA-associated renal vasculitis. *N Engl J Med* 363: 211–220, 2010. [PubMed: 20647198]

15. Jones RB, Furuta S, Cohen-Tervaert JW, Hauser T, Luqmani R, Morgan MD, et al.: Rituximab versus cyclophosphamide in ANCA-associated renal vasculitis: 2-year results of a randomised trial. *Ann Rheum Dis* 74: 1178–1182, 2015 [PubMed: 25739829]

16. Pusey CD, Rees AJ, Evans DJ, Peters DK, Lockwood CM: Plasma exchange in focal necrotizing glomerulonephritis without anti-GBM antibodies. *Kidney Int* 40: 757–763, 1991. [PubMed: 1745027]

17. Gaskin G, Pusey CD: Plasmapheresis in antineutrophil cytoplasmic antibody-associated systemic vasculitis. *Ther Apher* 5: 176–181, 2001. [PubMed: 11467753]

18. Jayne DR, Gaskin G, Rasmussen N, Abramowicz D, Ferrario F, Guillevin L, et al.; European Vasculitis Study Group : Randomized trial of plasma exchange or high-dosage methylprednisolone as adjunctive therapy for severe renal vasculitis. *J Am Soc Nephrol* 18: 2180–2188, 2007. [PubMed: 17582159]

19. Szpirt WM, Heaf JG, Petersen J: Plasma exchange for induction and cyclosporine A for maintenance of remission in Wegener's granulomatosis--a clinical randomized controlled trial. *Nephrol Dial Transplant* 26: 206–213, 2011. [PubMed: 20577017]

20. Walsh M, Casian A, Flossmann O, Westman K, Höglund P, Pusey C, et al.; European Vasculitis Study Group (EUVAS) : Long-term follow-up of patients with severe ANCA-associated vasculitis comparing plasma exchange to intravenous methylprednisolone treatment is unclear. *Kidney Int* 84: 397–402, 2013. [PubMed: 23615499]

21. Walsh M, Catapano F, Szpirt W, Thorlund K, Bruchfeld A, Guillevin L, et al.: Plasma exchange for renal vasculitis and idiopathic rapidly progressive glomerulonephritis: A meta-analysis. *Am J Kidney Dis* 57: 566–574, 2011. [PMCID: PMC3062650] [PubMed: 21194817]

22. Walters GD, Willis NS, Craig JC: Interventions for renal vasculitis in adults. A systematic review. *BMC Nephrol* 11: 12, 2010. [PMCID: PMC2914014] [PubMed: 20573267]

23. Frascà GM, Zoumparidis NG, Borgnino LC, Neri L, Neri L, Vangelista A, et al.: Combined treatment in Wegener's granulomatosis with crescentic glomerulonephritis--clinical course and long-term outcome. *Int J Artif Organs* 16: 11–19, 1993. [PubMed: 8458666]

24. de Joode AAE, Sanders JSF, Smid WM, Stegeman CA: Plasmapheresis rescue therapy in progressive systemic ANCAassociated vasculitis: Single-center results of stepwise escalation of immunosuppression. *J Clin Apher* 29: 266–272, 2014. [PubMed: 24497412]

25. Gregersen JW, Kristensen T, Krag SR, Birn H, Ivarsen P: Early plasma exchange improves outcome in PR3-ANCA-positive renal vasculitis. *Clin Exp Rheumatol* 30[Suppl 70]: S39–S47, 2012. [PubMed: 22510451]

26. Guillevin L, Fain O, Lhote F, Jarrousse B, Le Thi Huong D, Bussel A, et al.: Lack of superiority of steroids plus plasma exchange to steroids alone in the treatment of polyarteritis nodosa and Churg-Strauss syndrome. A prospective, randomized trial in 78 patients. *Arthritis Rheum* 35: 208–215, 1992. [PubMed: 1346499]

7/7/24, 3:08 PM Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody–Associated Vasculitis with Severe Kidney Disea...

27. Cole E, Cattran D, Magil A, Greenwood C, Churchill D, Sutton D, et al.: A prospective randomized trial of plasma exchange as additive therapy in idiopathic crescentic glomerulonephritis. The Canadian Apheresis Study Group. *Am J Kidney Dis* 20: 261–269, 1992. [PubMed: 1519607]

28. Glöckner WM, Sieberth HG, Wichmann HE, Backes E, Bambauer R, Boesken WH, et al.: Plasma exchange and immunosuppression in rapidly progressive glomerulonephritis: A controlled, multi-center study. *Clin Nephrol* 29: 1–8, 1988. [PubMed: 3289804]

29. Guillevin L, Lhote F, Cohen P, Jarrousse B, Lortholary O, Généreau T, et al.: Corticosteroids plus pulse cyclophosphamide and plasma exchanges versus corticosteroids plus pulse cyclophosphamide alone in the treatment of polyarteritis nodosa and Churg-Strauss syndrome patients with factors predicting poor prognosis. A prospective, randomized trial in sixty-two patients. *Arthritis Rheum* 38: 1638–1645, 1995. [PubMed: 7488285]

30. Guillevin L, Cevallos R, Durand-Gasselin B, Lhote F, Jarrousse B, Callard P: Treatment of glomerulonephritis in microscopic polyangiitis and Churg-Strauss syndrome. Indications of plasma exchanges, Meta-analysis of 2 randomized studies on 140 patients, 32 with glomerulonephritis. *Ann Med Interne (Paris)* 148: 198–204, 1997. [PubMed: 9255326]

31. Casian A, Jayne D: Plasma exchange in the treatment of Wegener's granulomatosis, microscopic polyangiitis, Churg-Strauss syndrome and renal limited vasculitis. *Curr Opin Rheumatol* 23: 12–17, 2011. [PubMed: 21124082]

32. Szpirt WM: Plasma exchange in antineutrophil cytoplasmic antibody-associated vasculitis--a 25-year perspective. *Nephrol Dial Transplant* 30: i146–i149, 2015. [PubMed: 25770166]

33. Lionaki S, Falk RJ: Removing antibody and preserving glomeruli in ANCA small-vessel vasculitis. *J Am Soc Nephrol* 18: 1987–1989, 2007. [PubMed: 17596633]

34. Walsh M, Merkel PA, Peh CA, Szpirt WM, Puéchal X, Fujimoto S, et al.; PEXIVAS Investigators : Plasma Exchange and glucocorticoids in severe ANCA-associated vasculitis. *N Engl J Med* 382: 622–631, 2020. [PMCID: PMC7325726] [PubMed: 32053298]

35. Stone JH, Hoffman GS, Merkel PA, Min YI, Uhlfelder ML, Hellmann DB, et al.; International Network for the Study of the Systemic Vasculitides (INSSYS) : A disease-specific activity index for Wegener's granulomatosis: Modification of the Birmingham vasculitis activity score. International Network for the Study of the Systemic Vasculitides (INSSYS). *Arthritis Rheum* 44: 912–920, 2001. [PubMed: 11318006]

36. Mukhtyar C, Lee R, Brown D, Carruthers D, Dasgupta B, Dubey S, et al.: Modification and validation of the Birmingham vasculitis activity score (version 3). *Ann Rheum Dis* 68: 1827–1832, 2009. [PubMed: 19054820]

37. Merkel PA, Cuthbertson DD, Hellmich B, Hoffman GS, Jayne DR, Kallenberg CG, et al.; Vasculitis Clinical Research Consortium : Comparison of disease activity measures for anti-neutrophil cytoplasmic autoantibody (ANCA)-associated vasculitis. *Ann Rheum Dis* 68: 103–106, 2009. [PMCID: PMC3139244] [PubMed: 18664546]

38. Jennette JC, Falk RJ, Bacon PA, Basu N, Cid MC, Ferrario F, et al.: 2012 revised international Chapel Hill consensus conference nomenclature of vasculitides. *Arthritis Rheum* 65: 1–11, 2013. [PubMed: 23045170]

39. Jennette JC, Falk RJ, Andrassy K, Bacon PA, Churg J, Gross WL, et al.: Nomenclature of systemic vasculitides. Proposal of an international consensus conference. *Arthritis Rheum* 37: 187–192, 1994. [PubMed: 8129773]

40. Leavitt RY, Fauci AS, Bloch DA, Michel BA, Hunder GG, Arend WP, et al.: The American College of Rheumatology 1990 criteria for the classification of Wegener's granulomatosis. *Arthritis Rheum* 33: 1101–1107, 1990. [PubMed: 2202308]

Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody–Associated Vasculitis with Severe Kidney Disea...

41. Levey AS, Stevens LA, Schmid CH, Zhang YL, Castro AF 3rd, Feldman HI, et al.; CKD-EPI (Chronic Kidney Disease Epidemiology Collaboration) : A new equation to estimate glomerular filtration rate [published correction appears in *Ann Intern Med* 150: 604–612, 2009. [PMCID: PMC2763564] [PubMed: 19414839]

42. Levey AS, Stevens LA: Estimating GFR using the CKD Epidemiology Collaboration (CKD-EPI) creatinine equation: More accurate GFR estimates, lower CKD prevalence estimates, and better risk predictions. *Am J Kidney Dis* 55: 622–627, 2010. [PMCID: PMC2846308] [PubMed: 20338463]

43. Padmanabhan A, Connelly-Smith L, Aqui N, Balogun RA, Klingel R, Meyer E, et al.: Guidelines on the use of therapeutic apheresis in clinical practice - evidence-based approach from the writing committee of the American society for apheresis: The eighth special issue. *J Clin Apher* 34: 171–354, 2019. [PubMed: 31180581]

44. Kidney Disease: Improving Global Outcomes (KDIGO) Glomerulonephritis Work Group: The 2020 clinical practice guidelines on glomerular diseases. 2020. Available at: <u>https://kdigo.org/guidelines/gn</u>. Accessed July 4, 2020

45. Hernán MA: The hazards of hazard ratios. *Epidemiology* 21: 13–15, 2010. [PMCID: PMC3653612] [PubMed: 20010207]

46. D'Agostino RB, Jr.: Propensity score methods for bias reduction in the comparison of a treatment to a non-randomized control group. *Stat Med* 17: 2265–2281, 1998. [PubMed: 9802183]

47. Hill J: Discussion of research using propensity-score matching: Comments on 'A critical appraisal of propensity-score matching in the medical literature between 1996 and 2003' by Peter Austin, Statistics in medicine. *Stat Med* 27: 2055–2061; discussion 2066–2069, 2008 [PubMed: 18446836]

48. Rhee RL, Hogan SL, Poulton CJ, McGregor JA, Landis JR, Falk RJ, et al.: Trends in long-term outcomes among patients with antineutrophil cytoplasmic antibody-associated vasculitis with renal disease. *Arthritis Rheumatol* 68: 1711–1720, 2016. [PMCID: PMC4920688] [PubMed: 26814428]

49. Binda V, Moroni G, Messa P: ANCA-associated vasculitis with renal involvement. *J Nephrol* 31: 197–208, 2018. [PubMed: 28560688]

50. Jennette JC, Nachman PH: ANCA glomerulonephritis and vasculitis. *Clin J Am Soc Nephrol* 12: 1680–1691, 2017. [PMCID: PMC5628710] [PubMed: 28842398]

51. Marco H, Draibe J, Villacorta J, Quintana LF, Martin N, Garcia-Osuna R, et al.; Spanish Group for the Study of Glomerular Disease (GLOSEN) : Determinants of renal and patient outcomes in a Spanish cohort of patients with ANCA-associated vasculitis and renal involvement. *Clin Rheumatol* 37: 1065–1074, 2018. [PubMed: 29520673]

52. Cartin-Ceba R, Diaz-Caballero L, Al-Qadi MO, Tryfon S, Fervenza FC, Ytterberg SR, et al.: Diffuse alveolar hemorrhage secondary to antineutrophil cytoplasmic antibody-associated vasculitis: Predictors of respiratory failure and clinical outcomes. *Arthritis Rheumatol* 68: 1467–1476, 2016. [PubMed: 26713723]

53. Specks U, Merkel PA, Hoffman GS, Langford CA, Spiera R, Seo P, et al.: Design of the rituximab in ANCA-associated vasculitis (RAVE) trial. *Open Arthritis Journal* 4: 1–18, 2011

Figures and Tables



Figure 1.



Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) flowchart for the selection of patients with severe renal involvement in AAV. Active renal involvement was defined by the presence of (1) active, biopsy-proven, pauciimmune GN; (2) red blood cell casts on urine microscopy; or (3) rise in SCr >30% (or >25% decline in creatinine clearance) attributed to active vasculitis. Severe kidney disease was defined by eGFR<30 ml/min per 1.73 m² at diagnosis of renal disease secondary to AAV. Patients were grouped according to the interventions received for remission-induction (CYC or RTX and with or without PLEX). To assess the efficacy of CYC (n=161) in comparison with RTX (n=64) for remission-induction in patients with AAV and severe renal involvement, we excluded 26 patients treated with other immunosuppressants. In the group of patients that received PLEX (n=51), 37 (72.5%) were treated with CYC and 14 (27.5%) were treated with RTX for remission-induction. EGPA, eosinophilic granulomatosis with polyangiitis; GPA, granulomatosis with polyangiitis.

Table 1.

Demographic and clinical characteristics of patients with AAV with active severe kidney disease

Characteristic	Active Severe Kidney Disease in AAV <i>n</i> =251
Age at diagnosis of severe renal involvement, median (IQR) yr ^a	66 (55-74)
Male, <i>n</i> (%)	128 (51.0)
Disease presentation, <i>n</i> (%)	
AAV new diagnosis	194 (77.3)
AAV relapse	57 (22.7)
AAV, n (%)	
MPA	140 (55.8)
GPA	111 (44.2)
ANCA specificity (ELISA), n (%)	
МРО	156 (62.2)
PR3	95 (37.8)
BVAS/WG at renal involvement diagnosis, median (IQR)	8 (7-10)
Renal limited disease, n (%)	84 (33.5)
Alveolar hemorrhage, n (%)	40 (15.9)
Pulmonary-renal syndrome, n (%)	33 (13.1)
Cardiovascular risk factors, n (%)	
Arterial hypertension	187 (74.5)
Diabetes mellitus	56 (22.3)
Dyslipidemia	91 (36.3)
BMI>30 kg/m ²	87 (34.7)
Laboratory findings	
Hemoglobin, mean (SD) g/dl	9.7 (8.7–11.0)
ESR>22 mm/h, n (%)	145 (57.8)
SCr at diagnosis, median (IQR) mg/dl	3.1 (2.5–4.2)
eGFR at diagnosis, mean (SD) ml/min per $1.73~{ m m}^2$	16.3 (10.2–21.9)
Biopsy proven, n (%)	222 (88.4)
Intervention	
Remission-induction treatment, n (%)	
CYC	161 (64 1)

EGFR is estimated using the Chronic Kidney Disease Epidemiology collaboration method. IQR, interquartile range; GPA, granulomatosis with polyangiitis; ESR, erythrocyte sedimentation rate.

^aTreatment was started on average within 16 h of the diagnosis of severe renal involvement.

Table 2.

Outcomes of patients with AAV with active severe kidney disease

Variable	Active Severe Kidney Disease in AAV n=251		
Outcomes			
Vasculitis, n (%)			
Remission			
6 mo	168 (66.9)		
Total ^a	198 (78.8)		
Time to remission, median (IQR) mo	3.7 (2.5–5.5)		
Complete remission			
At 6 mo	46 (18.3)		
Total ^a	119 (47.4)		
Time to complete remission, median (IQR) mo	11.9 (3.2–17.6)		
Renal remission			
At 6 mo	177 (70.5)		
Total ^a	199 (79.3)		
Relapse			
At 12 mo	32 (12.7)		
Total ^a	81 (32.3)		
Time to relapse, median (IQR) mo	16.0 (8.5–50.0)		
Renal relapse	46 (18.3)		
Death			
At 24 mo	22 (8.8)		
Total ^a	62 (24.7)		
Time to death, median (IQR) mo	44.8 (11.5-87.8)		
Death AAV related, n (%)	34 (13.5)		
Renal, <i>n</i> (%)			
ESKD			
At 24 mo	59 (23.5)		
Total ^a	78 (31.1)		
Time to ESKD, median (IQR) mo	3.0 (0.1–29.5)		

IQR, interquartile range; FU, follow-up.

^a"Total" refers to the number of occurrences during all follow-up time.

Figure 2.



Changes in mean eGFR with follow up. Estimates for the mean and SEM of eGFR (ml/min per 1.73 m^2) at baseline and at 6, 12, 18, and 24 months after beginning remission-induction therapy for each treatment group are displayed: (A) CYC versus RTX (mean eGFR at baseline, 15.6 versus 17.7 ml/min per 1.73 m^2 ; *P*=0.050); (B) PLEX versus no PLEX (mean eGFR at baseline, 14.1 versus 16.8 ml/min per 1.73 m^2 ; *P*=0.035); (C) CYC + PLEX versus RTX + PLEX (mean eGFR at baseline, 12.5 versus 18.5 ml/min per 1.73 m^2 ; *P*=0.026). GFR was estimated using the Chronic Kidney Disease Epidemiology collaboration formula.

Table 3.

Clinical characteristics of patients with AAV and active severe kidney disease treated with CYC or RTX for remission-induction (n=225)

Characteristic	CYC <i>n</i> =161	RTX <i>n</i> =64	Std. Diff.	
	(64.1%)	(25.5%)	(%) ^a	
Age at diagnosis of severe renal involvement, median (IQR) yr ^b	65 (55–74)	69 (59–75)	11.0	
Male, n (%)	86 (53.4)	32 (50.0)	-6.0	
AAV, n (%)				
MPA	83 (51.6)	35 (54.7)	6.2	
GPA	78 (48.4)	29 (45.3)	-6.2	
ANCA specificity (ELISA), <i>n</i> (%)				
MPO	95 (59.0)	38 (59.4)	0.8	
PR3	66 (41.0)	26 (40.6)	-0.8	
BVAS/WG at diagnosis, median (IQR)	8 (7-10)	8 (7-10)	0.0	
Renal limited disease at diagnosis, n (%)	50 (31.1)	21 (32.8)	3.6	
Rapidly progressive GN, n (%)	69 (42.8)	21 (26.6)	-34.5	
Alveolar hemorrhage BVAS/WG at diagnosis, n (%)	27 (16.8)	10 (15.6)	3.3 ^c	
Biopsy proven, n (%)	145 (90.1)	52 (81.3)	-25.3	
Cardiovascular risk factors, n (%)				
Arterial hypertension	119 (73.9)	48 (75.0)	2.5	
Diabetes mellitus	38 (23.6)	13 (20.3)	-8.0	
Dyslipidemia	59 (36.6)	26 (40.6)	8.2	
BMI>30 kg/m ²	51 (33.6)	27 (49.1)	31.8	
Laboratory findings				
Hemoglobin, mean (SD) g/dl	9.6 (8.6–10.7)	9.9 (8.9–11.1)	18.4	
ESR>22 mm/1st h, n (%)	88 (83.0)	39 (76.5)	-16.2	
eGFR at diagnosis of renal involvement, mean (SD) ml/min per $1.7\;\mathrm{3m^2}$	15.6 (7.2)	17.7 (6.9)	29.7	
eGFR at diagnosis of renal involvement <15 ml/min per 1.73 m ² , <i>n</i> (%)	71 (44.1)	24 (37.5)	-13.5 ^c	
eGFR at 6 mo, median (IQR)	28.8 (16.3-37.6)	30.5 (22.2-45.3)	13.4	
eGFR at 12 mo, median (IQR)	29.9 (20.5-41.5)	31.5 (19.2–44.7)	0.0	

eGFR is estimated using the Chronic Kidney Disease Epidemiology collaboration method. IQR, interquartile range; GPA, granulomatosis with polyangiitis; ESR, erythrocyte sedimentation rate. ^aStandardized mean differences.

7/7/24, 3:08 PM Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody–Associated Vasculitis with Severe Kidney Disea...

^bTreatment was started on average within 16 h of the diagnosis of severe renal involvement.

^cVariables used in the propensity score.

Table 4.

Outcomes of patients with AAV and active severe kidney disease treated with CYC or RTX for remission-induction (*n*=225)

Variable	Prematching			Postmatching ^a		
	CYC <i>n</i> =161	RTX <i>n</i> =64	Р	CYC <i>n</i> =60	RTX <i>n</i> =60	Р
	(64.1%)	(25.5%)	Value ^b	(37.3%)	(93.8%)	Value ^b
Outcomes						
Vasculitis activity, n (%)						
Remission						
6 mo	105 (76.6)	50 (83.3)	0.291	45 (75.0)	49 (81.7)	0.375
Total	124 (87.9)	54 (87.1)	0.866	53 (93.0)	54 (91.5)	0.769
Complete remission						
6 mo	23 (17.7)	19 (31.7)	0.031	12 (25.0)	18 (31.6)	0.457
Total	72 (54.5)	38 (66.7)	0.178	31 (54.4)	37 (62.7)	0.096
Renal remission						
6 mo	110 (78.0)	52 (83.9)	0.338			
Total	125 (88.7)	54 (87.1)	0.752			
Relapse						
Total	52 (38.5)	23 (40.4)	0.812			
Renal relapse	26 (16.1)	16 (25.0)	0.135			
Death						
24 mo	13 (8.1)	8 (12.5)	0.303			
Total	47 (29.2)	12 (18.8)	0.108			
Renal recovery, <i>n</i> (%)						
ESKD						
24 mo	45 (28.0)	10 (15.6)	0.052			
Total	58 (36.0)	11 (17.2)	0.006			
Dialysis (total)	41 (25.5)	4 (6.3)	0.002			
Renal function recovery after 6 mo to an eGFR>30 ml/min per 1.73 m ²	47 (46.5)	22 (51.2)	0.611			
Combined events of ESKD and/or death, <i>n</i> (%)						
		. =				

Population follow-up: 6 mo, CYC: *n*=137 versus RTX: *n*=57; 18 mo, CYC: *n*=120 versus RTX: *n*=51; 24 mo, CYC: *n*=112 versus RTX: *n*=49. IQR, interquartile range; FU, follow-up.

^aThe pairs used for the variables comparison after matching were obtained by analyzing the outcome of remission at 6 mo. Therefore, the comparisons pertaining to other outcomes are not shown.

7/7/24, 3:08 PM Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody–Associated Vasculitis with Severe Kidney Disea...

^b*P* value <0.05 is considered significant (Pearson chi-squared test for categoric variables, *t* test for continuous variables with normal distribution, and Mann–Whitney *U* test for continuous variables with skewed distribution). ^c"Total" refers to the number of occurrences during all follow-up time.

Figure 3.



Remission and combined events after PS matching analysis. Kaplan–Meier plots of remission achieved over 6 months after initiating remission-induction therapy ([A] 60 CYC versus 60 RTX patients, 45 versus 49 events, time to event 4.1 versus 3.5 months, *P*=0.069), and combined events (ESKD and/or death) at 18 months ([B] 64 CYC versus 64 RTX patients, 16 versus 14 events, time to event 14.3 versus 14.7 months, *P*=0.698), according to the remission-induction immunosuppression, CYC versus RTX.

Table 5.

Risk for the outcome stratified by treatment after PS matching analysis

Variable	Post-PS Matching Analysis			
	Logistic Regression		Cox Regression	
	OR (95% CI)	P Value	IRR (95% CI)	P Value ^a
Remission at 6 mo				
RTX versus CYC	1.49 (0.62 to 3.57)	0.377	1.37 (0.91 to 2.08)	0.132
PLEX versus no PLEX	0.61 (0.23 to 1.63)	0.323	0.73 (0.44 to 1.22)	0.230
Relapse at 12 mo				
RTX versus CYC	1.81 (0.69 to 4.80)	0.230	1.61 (0.67 to 3.89)	0.289
PLEX versus no PLEX	1.90 (0.51 to 7.09)	0.337	0.53 (0.16 to 1.82)	0.315
Combined events of ESKD and/or death at 18 mo				
RTX versus CYC	0.84 (0.37 to 1.91)	0.677	0.87 (0.42 to 1.78)	0.699
PLEX versus no PLEX	1.11 (0.46 to 2.68)	0.822	1.05 (0.51 to 2.18)	0.891
Survival at 24 mo				
RTX versus CYC	1.69 (0.52 to 5.46)	0.384	1.57 (0.51 to 4.80)	0.429
PLEX versus no PLEX	0.53 (0.14 to 1.95)	0.340	0.54 (0.16 to 1.85)	0.329
ESKD at 24 mo				
RTX versus CYC	0.56 (0.23 to 1.34)	0.191	0.61 (0.28 to 1.35)	0.224
PLEX versus no PLEX	1.11 (0.45 to 2.74)	0.818	1.06 (0.50 to 2.25)	0.887
Combined events of ESKD and/or death at 24 mo				
RTX versus CYC	0.72 (0.32 to 1.60)	0.415	0.77 (0.38 to 1.54)	0.452
PLEX versus no PLEX	1.00 (0.42 to 2.40)	1.000	0.99 (0.48 to 2.02)	0.971

 ^{a}P value <0.05 is considered significant.

Table 6.

Clinical characteristics of patients with AAV and active severe kidney disease according with the status of treatment with PLEX (*n*=251)

Characteristic	No PLEX <i>n</i> =200	X n=200 PLEX n=51	
	(79.7%)	(20.3%)	(%)"
Age at diagnosis of severe renal involvement, median (IQR) $$\rm yr^{b}$$	67 (55–75)	65 (56–74)	-3.5
Male, <i>n</i> (%)	97 (48.5)	31 (60.8)	24.9
AAV, n (%)			
MPA	120 (60.0)	20 (39.2)	-42.5
GPA	80 (40.0)	31 (60.8)	42.5
ANCA specificity (ELISA), n (%)			
MPO	134 (67.0)	22 (43.1)	-49.5
PR3	66 (33.0)	29 (56.9)	49.5
BVAS/WG at diagnosis, median (IQR)	7 (7–10)	9 (7-12)	39.7
Renal limited disease at diagnosis, n (%)	74 (37.0)	10 (19.6)	-39.3
Rapidly progressive GN, n (%)	74 (37.0)	23 (45.1)	16.5
Alveolar hemorrhage BVAS/WG at diagnosis, n (%)	25 (12.5)	15 (29.4)	42.4 ^c
Biopsy proven, n (%)	178 (89.0)	44 (86.3)	-8.2
Cardiovascular risk factors, <i>n</i> (%)			
Arterial hypertension	151 (75.5)	36 (70.6)	-11.1
Diabetes mellitus	43 (21.5)	13 (25.5)	9.4
Dyslipidemia	71 (35.5)	20 (39.2)	7.7
BMI>30 kg/m ²	67 (36.0)	20 (44.4)	17.2
Laboratory findings			
Hemoglobin, mean (SD) g/dl	9.9 (9.0-11.3)	9.2 (7.9–10.0)	-64.6
ESR>22 mm/1st h, n (%)	122 (81.9)	23 (79.3)	-6.6
eGFR at diagnosis of renal involvement, mean (SD) ml/min per 1.73 m ²	16.8 (7.0)	14.1 (8.0)	-36.0
eGFR at diagnosis of renal involvement <15 ml/min per 1.73 m ² , n (%)	75 (37.5)	29 (56.9)	39.6 ^c
eGFR at 6 mo, median (IQR)	29.9 (20.1-40.5)	27.7 (14.2-43.0)	-10.1
eGFR at 12 mo, median (IQR)	31.8 (21.2-42.0)	30.6 (22.1-49.2)	0.5

eGFR is estimated using the Chronic Kidney Disease Epidemiology collaboration method. IQR, interquartile range; GPA, granulomatosis with polyangiitis; ESR, erythrocyte sedimentation rate. ^aStandardized mean differences.

7/7/24, 3:08 PM Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody–Associated Vasculitis with Severe Kidney Disea...

^bTreatment was started on average within 16 h of the diagnosis of severe renal involvement.

^cVariables used in the propensity score.

Table 7.

Outcomes of patients with AAV and active severe kidney disease according with the status of treatment with PLEX (*n*=251)

	No DI EV				Postmatching ^a		
	norlex n=200 (79.7%)	PLEX n=51 (20.3%)	<i>P</i> Value ^b	No PLEX n=45 (22.5%)	PLEX n=45 (88.2%)	<i>P</i> Value ^b	
Outcomes							
Vasculitis activity, n (%)							
Remission							
6 mo	132 (78.1)	36 (73.5)	0.497				
Total ^c	155 (88.6)	43 (87.8)	0.875				
Complete remission							
6 mo	35 (21.7)	11 (23.9)	0.754				
Total ^c	91 (52.0)	28 (57.1)	0.516				
Renal remission							
6 mo	139 (79.4)	38 (77.6)	0.775				
Total ^c	155 (88.6)	44 (89.8)	0.810				
Relapse							
12 mo	25 (13.1)	7 (14.9)	0.745				
Total ^c	62 (37.8)	19 (38.8)	0.902				
Renal relapse	36 (59.0)	10 (52.6)	0.623				
Death							
24 mo	18 (9.0)	4 (7.8)	0.794				
Total ^c	45 (22.5)	17 (33.3)	0.109				
Renal recovery, <i>n</i> (%)							
ESKD							
24 mo	42 (21.0)	17 (33.3)	0.064				
Total ^c	57 (28.5)	21 (41.2)	0.081				
Dialysis (total)	34 (17.0)	20 (39.3)	< 0.0001	7 (15.5)	16 (35.5)	0.087	
Renal function recovery after 6 mo to an eGFR>30 ml/min per 1.73 m ²	61 (50.0)	17 (44.7)	0.571				

Population follow-up: 6 mo, no PLEX: *n*=172 versus PLEX: *n*=45; 12 mo, no PLEX: *n*=157 versus PLEX: *n*=45; 18 mo, no PLEX: *n*=147 versus PLEX: *n*=43; 24 mo, no PLEX: *n*=142 versus PLEX: *n*=39. IQR, interquartile range; FU, follow-up. ^aThe pairs resultant from the PS matching were obtained by the analysis of the outcome of combined events of ESKD and/or death at 18 mo. Therefore, the comparisons pertaining to other outcomes are not shown.

7/7/24, 3:08 PM Efficacy of Rituximab and Plasma Exchange in Antineutrophil Cytoplasmic Antibody–Associated Vasculitis with Severe Kidney Disea...

^b*P* value <0.05 is considered significant (Pearson chi-squared test for categoric variables, *t* test for continuous variables with normal distribution, and Mann–Whitney *U* test for continuous variables with skewed distribution). ^c"Total" refers to the number of occurrences during all follow-up time.

Figure 4.



Remission and combined events after PS matching analysis. Kaplan–Meier plots of remission achieved over 6 months after initiating remission-induction therapy ([A] 45 no PLEX versus 45 PLEX patients, 34 versus 30 events, time to event 4.0 versus 4.1 months, P=0.176), and combined events (ESKD and/or death) at 18 months ([B] 45 no PLEX versus 45 PLEX patients, 14 versus 15 events, time to event 13.3 versus 13.0 months, P=0.891), according to the status of treatment or no treatment with PLEX.